Author's response to reviews

Title: A randomised, controlled crossover comparison of the C-MAC videolaryngoscope with direct laryngoscopy in 150 patients during routine induction of anaesthesia

Authors:

Erol Cavus (cavus@anaesthesie.uni-kiel.de)
Carsten Thee (thee@anaesthesie.uni-kiel.de)
Thora Moeller (thora.moeller@klinik sued-rostock.de)
Joerg Kieckhaefer (joerg.kieckhaefer@klinik sued-rostock.de)
Volker Doerges (doerges@anaesthesie.uni-kiel.de)
Klaus Wagner (klaus.wagner@klinik sued-rostock.de)

Version: 7 Date: 11 February 2011

Author's response to reviews:

Dear Editorial team,

we have changed the manuscript according to your specifications and again have carefully read and corrected the manuscript, if necessary.